

EO-1 Weekly Report


	Earth Observing – 1

	December 22, 2011 – January 4, 2012

	DOY
	356
	–
	004

	Mission Day
	4109
	–
	4122


EO-1 SPACECRAFT

Instruments

· Scheduled 155 science Data Collection Events (DCEs) the first week 

· Received all images
· Scheduled 142 science Data Collection Events (DCEs) the second week 

· Received all images
· Performed instrument decontamination cycles

· Conducted HSI deicing from 002/00:25z to 002/15:15z

· Conducted ALI outgassing from 002/00:35z to 002/15:25z
· Performed Instrument calibration

· Ali Internal Calibration II at 360/00:45z

· Solar Array Characterization at 002/10:19z

· HSI solar calibration 003/15:53z
EO-1 MISSION OPERATIONS CENTER

Real-Time

· Maintained nominal spacecraft state of health
· Setting up and performing testing and engineering passes with TrollSat ground station

· Completed another set of TR2 passes nominally

· TrollSat (TR2) is now certified for S-Band passes and will be putting 2-3 per week in the schedule
· Had an issue with SGS where we did not get receiver or command lock for most of the pass

· Downlink still came in fine, the ground station was using an incorrect uplink frequency, this has been corrected

· Monitoring the battery differential voltage and Solar Array Temperatures
· The battery differential voltage has hit -.16 volts which is tripping a TSM that gives a warning message, this is a little less than half of the voltage needed to trip the TSM that actually takes action

· The total voltage, current, temperature and end of day/night are all reading normal which suggests that either a cell on the bottom half is charging fast or one on the upper half is charging slow

· Working with Power people to determine the best course of action to take

· If we determine a cell is charging slower, raise the VT level, and if one is charging faster, lower the VT level

· Putting the day 009 Lunar calibration on hold until we determine the action to resolve the differential voltage

· Monitoring SGS communications link problem

· There is a work around in place until the cables are repaired.

· Adds a small latency to commands and command response

· This latency is acceptable

· Continuing to monitor SGS passes to ensure they remain stable
· SGS has an equipment problem and will be down until approximately August 2012
· SGS passes being taken on the SG-1/SG-22 hybrid antenna that was used during last equipment malfunction

· Looking into testing/certifying SG2 in X-band supports (currently it is certified for S-band only supports)

· SG2 is shadowing our supports and will be transferring the data to us to test X-Band data

Mission Planning
· Looking at the steps required in creating a South Pole image

· Creating the Lunar Calibration products for the day 009 calibration in case we end up doing it

· Performed the mission planning activities for generating TrollSat supports

· Remaking the SOP for mission planning activities for ASPEN and ASIST SCP
· Creating and modifying scripts to help perform various mission planning activities

· ASPEN (both UDAP and daily file)

· Verifying
· Lunar Calibration

· TDRS 

· The TDRS daily file generation was modified due to the real time script marking them as handover

· The modifications are being verified and they work in real time as well as data processing

· Weekly Supports

· ASIST SCP

· Finalizing

· Lunar Calibration

· Outgassing
· Verifying

· Backup Load Generation

· Load XML processing

· Load Checksum

Flight Dynamics
· Looking at the steps required to take a South Pole image

· Creating the Lunar Calibration products for the day 009 calibration in case we do go through with it

· Encountered a problem with the year rollover with Matlab trying to use dates not in a specified range

· Working to correct this

· Worked with FDF for getting products from their new Product Servers

· Products needed for operations are on the new server and can be retrieved

· They are still setting up their deliveries to the locations, but once they are established we will be able to move automation to the new product server

· Modifying FD scripts to get the FDF products through SFTP from the new Product Server

· Scripts have been modified and will work once a setting is changed on the FDF side

· Making Waldorf2 an active backup of Statler2
· Deliveries to Statler2 are also being made to Waldorf2
· Getting necessary programs/utilities set up on Waldorf2
· Copied the structure of Statler2 into Waldorf2 

· Added routine scheduled tasks to Waldorf2

· Disabled FDF product email capability on Waldorf2 (will be enabled if prime)

· Disabling FDF product delivery on Waldorf2 (will be enabled if prime)
MOC Move
· The last few cables for the mission monitors need to be run
System Administration
· Completed applying the Nov and Dec 2011 updates/patches on the 6 EO-1 CNE computers. The patches include Adobe Reader, Java, Firefox, and Microsoft products.

· Reviewing system logs for the period of time during my vacation/holidays. 

GROUND AND SPACE NETWORK

Station Downtimes 

· SGS has had an equipment malfunction and latest indication is that it will be back up next August
· SG1/22 has a communication line problem, workaround is in place, fix timing unknown
Operational Discrepancies

· None
Number of Real-Time Supports

[image: image1.emf]Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

22-Dec-11 356

7 1 0 0

8

23-Dec-11 357

5 3 1 0

9

24-Dec-11 358

6 3 0 0

9

25-Dec-11 359

7 4 0 0

11

26-Dec-11 360

5 4 0 0

9

27-Dec-11 361

5 3 0 0

8

28-Dec-11 362

5 4 1 0

10

40 22 2 0 64 Weekly Totals



[image: image2.emf]Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

29-Dec-11 363

5 4 1 0

10

30-Dec-11 364

6 4 0 0

10

31-Dec-11 365

6 3 1 0

10

1-Jan-12 1

6 3 1 0

10

2-Jan-12 2

3 7 0 0

10

3-Jan-12 3

5 4 0 0

9

4-Jan-12 4

6 3 0 0

9

37 28 3 0 68 Weekly Totals


UPCOMING EVENTS

· Next lunar calibration January 9, 2012
· Migrating the mission clock monitor cables to the new MOC
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		22-Dec-11		356		7		1		0		0		8

		23-Dec-11		357		5		3		1		0		9

		24-Dec-11		358		6		3		0		0		9

		25-Dec-11		359		7		4		0		0		11

		26-Dec-11		360		5		4		0		0		9

		27-Dec-11		361		5		3		0		0		8

		28-Dec-11		362		5		4		1		0		10

		Weekly Totals				40		22		2		0		64
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Sheet1

		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		29-Dec-11		363		5		4		1		0		10

		30-Dec-11		364		6		4		0		0		10

		31-Dec-11		365		6		3		1		0		10

		1-Jan-12		1		6		3		1		0		10

		2-Jan-12		2		3		7		0		0		10

		3-Jan-12		3		5		4		0		0		9

		4-Jan-12		4		6		3		0		0		9

		Weekly Totals				37		28		3		0		68
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